Author's response to reviews

Title: Burden of micronutrient deficiencies by socio-economic strata in children aged 6 months to 5 years in the Philippines

Authors:

Simon Wieser (wiso@zhaw.ch)
Rafael Plessow (plet@zhaw.ch)
Klaus Eichler (eich@zhaw.ch)
Olivia Malek (mals@zhaw.ch)
Urs Bruegger (brgu@zhaw.ch)

Version: 2 Date: 28 February 2013

Author's response to reviews:

The editor has asked us to add an ethics statement and to add a statement on whether the databases used are freely available:

(1) Our research does not require any ethical approval as it is entirely based on secondary analysis of primary data, as in the case of systematic reviews.

(2) We have added a statement on the freedom of access to the databases used in the study on cover letter.